Author's response to reviews

Title: Molecular evidence of Ureaplasma urealyticum and Ureaplasma parvum colonization in preterm infants during respiratory distress syndrome.

Authors:

Rosario Cultrera (ctr@unife.it)
Silva Seraceni (ctr@unife.it)
Rossella Germani (fdivalentin@candy.it)
Carlo Contini (cnc@unife.it)

Version: 7 Date: 7 November 2006

Author's response to reviews:

Ferrara, 07/11/2006
To The Chief Editor of
BMC Infectious Diseases

Dear Editor,

We would like to submit the revised manuscript entitled "Molecular evidence of Ureaplasma urealyticum and Ureaplasma parvum colonization in preterm infants during respiratory distress syndrome", by R. Cultrera et al., for publication in BMC Infectious Diseases. Authors considered the general comments indicated by BioMed Central Editorial board and the comments of the Reviewers.

You will find enclosed the figures and tables. I ensure that the manuscript was conform to the required format. This paper is original and has not been submitted in equivalent form for publication elsewhere. All Authors have read and approved this revised version and they have contributed significantly to the work. We declare there are no competing financial or non-financial interests (political, personal, religious, ideological, academic, intellectual, commercial or any other) in relation to this manuscript.

With best regards,
Rosario Cultrera

Corresponding Author:
Rosario Cultrera, M.D.
Infectious Diseases,
Department of Clinical and Experimental Medicine,
University of Ferrara
Via Fossato di Mortara, 23
44100 Ferrara, Italy
tel./fax +39 0532 291308, e-mail: ctr@dns.unife.it

Alternate Author:
Carlo Contini, M.D.
Infectious Diseases,
Department of Clinical and Experimental Medicine,
University of Ferrara
Via Fossato di Mortara, 23
44100 Ferrara, Italy
tel./fax +39 0532 291310, e-mail: cnc@dns.unife.it

ANSWERS TO THE REVIEWERS' COMMENTS

RESEARCH ARTICLE - MS: 1312995944104563

Molecular evidence of Ureaplasma urealyticum and Ureaplasma parvum colonization in preterm infants during respiratory distress syndrome.
Reviewer 3 (Prof. Arnold Smith)

ANSWERS

1. Spelling and grammatical errors. The manuscript has been reviewed one more time by an English translator.